Author’s response to reviews

Title: Unusual pathological fracture of the clavicle revealing primary hyperparathyroidism: a case report

Authors:

Yassir Benameur (benameur.yassir@gmail.com)
Hasnae Guerrouj (guerroujh@yahoo.fr)
Imad Ghfir (i.ghfir0@yahoo.fr)
Nouzha Ben Raïs Aouad (benraisaouadn@gmail.com)

Version: 2 Date: 06 Nov 2017

Author’s response to reviews:

Dear Reviewers;

I added on the manuscript a paragraph at the end of the introduction, to explain the importance of the case report in the medical knowledge.

All the informations about the patient, details of his physical examination on admission, and results of laboratory findings were also added.

The follow-up was mentionned at the end of the case presentation section.

On the discussion section, I started with summarizing the case, and its interest for the clinician.

The changes were highlighted in red colour on the manuscript.

Regards